

How does context influence players' behaviour? Experimental assessment in a 3-player coordination problem

Mathieu Desole, Stefano Farolfi, Patrick Rio

▶ To cite this version:

Mathieu Desole, Stefano Farolfi, Patrick Rio. How does context influence players' behaviour? Experimental assessment in a 3-player coordination problem. 2012. hal-02807029

HAL Id: hal-02807029 https://hal.inrae.fr/hal-02807029

Preprint submitted on 6 Jun 2020

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers.

L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

DOCUMENT de RECHERCHE

 How does context influence players' behaviour?
 Experimental assessment in a 3-player coordination problem »

Mathieu Désolé Stefano Farolfi Patrick Rio

DR n°2012-36

Unite de Formation et de Recherche d'Economie Avenue Raymond DUGRAND C.S. 79606 3 4 9 6 0 MONTPELLIER Cedex 2

Laboratoire Montpelliérain d'Economie Théorique et Appliquée

Unité Mixte de Recherche

E-mail : lameta@lameta.univ-montp1.fr web : www.lameta.univ-montp1.fr









How does context influenceplayers' behaviour? Experimental assessment in a 3-player coordination problem

Mathieu Désolé^a, Stefano Farolfi^{b,c}, Patrick Rio^d

^aMontpellier SupAgro, UMR LAMETA, F-34000 Montpellier, France

^bCIRAD, UMR G-EAU, F-34398 Montpellier, France

^cUniv. E Mondlane, International Centre for Water Economics& Governance Africa (IWEGA), Maputo, Mozambique

^d INRA, UMR LAMETA, Montpellier, F-34000 Montpellier, France

Abstract

This paper uses a CGT TU game modified into a coordination experiment to explore the causal effect of context on players' behaviour. An analytical framework focusing on four attributes representative of the game's context is proposed and an experimental protocol based on this framework allows testing hypotheses regarding the influence of context on players' choices. Results show that attributes such as Repetition and Communication seem to have a higher influence than Illustration on players' behaviour. The peculiar nature of the experimental results in the control group, showing the emergence of a focal point other than the outcome prescribed by the theory, allows discussing the expected "noise" observed in the treatments from a new perspective.

Key words: Context, Experimental Economics, Coordination, Cooperative Game Theory, Schelling effect, Role Playing Games, Water.

JEL: C92 ; C71 ; D71 ; Q25

Introduction

A common issue in experimental economics (EE) is the trade-off between internal and external validity of the experiments. Campbell (1957) considered an experiment internally valid if the experimenter finds a significant difference between the treatment and the control conditions. This difference is only the result of the response of subjects to causal processes at play and excludes all extraneous mediating factors that influence such responses. On the other hand, external validity is a matter of generalization of experimental results outside the specific situation created in the laboratory (Druckman et al., 2012).

Experiments conducted by economists tend to give priority to internal validity, by creating the laboratory conditions of control and replicability necessary to "evaluate the causal impacts of potentially informative explanatory variables" (Druckman et al., 2012). These conditions imply a simplification of the experimental settings, making them very distant from the real life, and therefore creating a problem in terms of external validity of the experiments (i.e. generalization of the outcomes).

Conversely, in the field of "adaptive and participatory research" (Barreteau et al., 2011), role-playing games (RPGs) are an "extremely simplified representation of the problem, yet sufficient to reflect the complexity of the system by taking the main dynamics and interactions into account" (Bousquet et al., 2011). As a consequence, protocols are highly contextualised. Obviously, RPGs are weak in terms of internal validity, as their complexity makes it difficult to interpret the results and identify the factors influencing the behaviour observed during the game (Desolé et al., 2009). Another problem with RPGs used as participatory platforms resides in their specificity to the local situation where they have been constructed, making them unsuitable for other similar utilisations (out-scaling) and consequently hampering the accumulation of knowledge that could result from this generalization (Rouchier, 2007).

Field experiments offer an intermediate set-up, where internal validity is preserved through rigorous protocols and control of the explanatory variables, but in order to increase external validity, the experiments are conducted far from laboratories, in the presence of more real conditions and with real stakeholders.

According to Shadish et al. (2002), "external validity covers at least four aspects of experimental design: whether the participants resemble the actors who are ordinarily confronted with these stimuli; whether the context (including the time), within which actors operate, resembles the context (and time) of interest; whether the stimulus used in the study resembles the stimulus of interest in the world; and whether the outcome measures resemble the actual outcomes of theoretical and practical interest".

We concentrated our analysis on the context of the experiment, which was identified with the level of information given to players through instructions, exchanged by players during an experiment or accumulated by players before or during an experiment.

The trade-off between internal and external validity of experiments assumes a different weight in function of the social discipline where these experiments are conducted. "While psychologist pay primary attention to issues associated with internal validity, political scientists tend to focus, almost exclusively, on problems associated with external validity" (Mc Dermott, 2012).

Limiting our analysis to EE and RPGs for participatory research, we observe that on one hand experimental economists are looking for more external validity in order to generalize the results of their abstract and controlled protocols, while on the other hand, although primarily driven by social validation (Le Page et al., 2011), RPGs designers are in quest of improved internal validity and therefore a higher capacity to interpret the games' outcomes and the factors that affect players' behaviour.

If we identify then a positive relation between the quantity of context (i.e. more realistic and informed set-up of a protocol) and the external validity of an experiment, we observe a tendency to contextualize abstract experiments in EE. At the same time, some RPGs designers (Ferrand et al., 2009; Abrami et al., 2012) tend to simplify their platforms to face the problems of interpretation and accumulation of knowledge indicated byDesolé et al. (2009) andRouchier (2007).

A crucial scientific issue is therefore the research of the quantity and quality of context that are "good" for both experimental protocols and RPGs in order to reach the acceptable compromise between internal and external validity in both disciplines and to allow simplification towards some accumulation of knowledge in RPGs.

A research programme that aims at answering this question consists for EE in finding out first how the context, as the previously defined array of variables, has a causal impact on player's behaviours, and then whether or not it is possible to isolate this impact and measure it quantitatively and objectively.

The experimental observations derived from such a research programme would provide useful suggestions and guidance for both economists and RPG designers in terms of the priorities to follow when choosing the contextual components constituting an experimental protocol or a game platform.

The idea developed in this paper is that experimentation makes it possible to analyse the question of the influence of context on players' behaviour.

More precisely, we are interested in characterising and quantifying the causal impact that introducing different dimensions of a predefined context in a completely decontextualized protocol has on the experiment's outcomes.

In other terms: do contextualized treatments, introducing more variables that influence players' behaviour, create a *noise* (Babcock et al., 1995) that increases the dispersion of players' responses if compared to the abstract control, or vice versa may the context rather be seen as a catalyst accelerating the learning processes without modifying the cognitive processes of reasoning (Cooper and Kagel, 2003)?

To answer our research question, we proceeded through the following steps.

First we proposed a new analytical framework for the definition and study of the contextual dimensions of an EE set-up. Then we designed an experimental protocol derived from the degradation of a role-playing game called "KatAware", created and used in South Africa to deal with decision-making about the allocation of water resources (Farolfi and Rowntree, 2007) in a catchment area. Our protocolconsists of a coordination problem where three players must choose to allocatethe grand-coalition payoff following the Cooperative Game Theory (CGT) theoretical reference.Finally, we used the analytical framework to build contextualized treatments in order to test our hypotheses about the impact of context on players' behaviour.The hypotheses that we test are therefore on the gap that the introduction of contextual elements creates between the outcomes of the untreated state (abstract) and the outcomes of the contextualized treatments.

The paper is structured as follows. Section 1 describes the new analytical framework to define and analyse the contextual dimensions of an experimental protocol; section 2 presents the coordination problem and section 3 illustrates how the coordination problem was converted into an experimental protocol aiming at testing hypotheses about the influence of context on players' behaviour. Section 4 presents and discusses the results of the experiments. Section 5 concludes, indicates the limits and caveats of the exercise and provides perspectives for future work.

1. A framework for the analysis of context in economic experiments

1.1 Context and players' behaviour

In endeavouring to obtain better control of the experiment parameters, investigators in EE agree that it is necessary to eliminate every element that might disrupt participants' behaviour. In light of the loss of control over behaviour caused by the different representations envisaged by the players, the introduction of context in experiments can be perceived as *noise* (Babcock et al., 1995). The underlying idea is that the participants have the same interpretation of a neutral and generic piece of information, hence the desire for maximum control to ensure the validity and robustness of the results.

Nevertheless, this idea of a homogenous perception of a neutral context by the players is called into question. According to certain authors, introducing references to reality has the effect of limiting players' confusion with regard to the decisions to be taken and the actions to be implemented. In other words, players can lose their ability to reason when they are placed in an environment which is too abstract and which, by definition, is unfamiliar to them as they do notencounter such abstract contexts in reality (Loewenstein, 1999;Loomes, 1999). If the subjects understand neither the question posed, by means of the instructions, nor the consequences of the decisions they are to take, no control over the players' motivations is guaranteed (Pillutla and Chen, 1999).

Faravelli (2007) compared the results obtained by conducting an experiment involving different samples of students enrolled in economics or sociology courses, either in the first year or subsequent years of their study course. He noted that the differences between the answers observed in abstract treatments disappear once context has been introduced, thereby corroborating the hypothesis made by Loewenstein (1999). The context provided players with indicators enabling them to behave in accordance with a common representation of this context. The players gave the same answer irrespective of their individual characteristics (level of studies and specialist subject), thereby contradicting – through the homogenisation of their behaviours – the idea that context acts as a *noise* which alters the interpretation of the economic question asked.

Cooper and Kagel (2003) identified two levels of contextual influence on behaviour. They adopted the hypothesis of a first "weak" level where context may be seen as a catalyst, accelerating the learning processes (established as the periods are repeated) without modifying the cognitive processes of reasoning. This "weak" level differs from the second "strong" level where context influences both the learning processes and the cognitive processes of decision-making, altering behaviours and encouraging much more sophisticated reasoning than that observed in abstract situations. Cooper and Kagel (2003) concluded that introducing contextual elements into protocols by means of a "weak" contextual effect is likely to allow behaviours to converge more quickly on the equilibrium observed in the protocols where these contextual elements do not exist.

1.2 Breaking context down into fourattributes

The literature presented above highlights the importance to experiments of the effect of context on participants' behaviour. The decisions taken by the players are based on social and ethical representations of the problems submitted to them. These representations depend on the context in which the agents are placed, i.e. the information available to them. Certain authors, such as Loewenstein (1999), recommend not attempting to eliminate every contextual reference in the protocols, instead endeavouring to understand the influence of context on behaviour. To this end, investigators shouldgo beyond the usual methodological standards in constructing their protocol in order to reflect reality more accurately, either by giving the players the ability to communicate (Cooper et al., 1989), or by selecting samples of players which are more representative of society

(Harrison and List, 2004), or by introducing a less abstractinformational content in the instructions given to the players (Wang, 1996).

The assessment of the influence of each of these different elements on the players' representationsis likely offer a better understanding of the decision-making processes in an experimental situation. Animproved knowledge of the effects produced by the introduction of contextual elements helps to target the construction of contextualised protocols more effectively and thus to improve the quality of experimental results, particularly their external validity, i.e. the capacity to extrapolate the observations made in the experimental conditions with regard to reality (Loewenstein, 1999).

In EE, the context is more often than not reduced to the information provided in the instructions and its effect is limited to the "framing effect", the definition of which resulting from the works of Tversky and Kahneman (1981) is that a minimal change in the description of the same decision-making problem can lead to a reversal of the players' preferences.

To answer our research question about the influence of context on players' behaviour, we argue that it is relevant not to limit the definition of a game context to this single aspect, but rather to consider it more broadly in terms of information conveyed in the protocol and characterised by:

- **the information that players accumulate during the game session** referring to the repetition of periods in a session and the subsequent*learning effect* that this can produce on players;
- **the information provided to the players** through the game instructions, or information flows from the scientist to the players;
- **the information exchanged among the players** by means of communication, or shared information;
- **the information already in possession of the players** referring to the players' experience.

Scientists exercise considerable control over the information given to the players, and by limiting the periods of a session to one (one-shot games) or few they can reduce to the minimum the *learning effect*. The information in possession of the players can be partially controlled by the selection of homogeneous groups of participants (e.g. students) to a game session, while the information exchanged among players can be observed through methods that constrain communication (cf. below).

Breaking down the context of a game, interpreted as a level of information, enables us to identify four elementary dimensions or attributes:

- 1. The first attribute- "REPETITION" takes into account the progression of the periods which contributes to the *learning process*, i.e. the accumulation of information during the game leading to increased knowledge. By accumulating information, a player can change his behaviour in order to adapt to the characteristics of the game and the decisions taken by his partners (Roth and Erev, 1995; Hertwig and Ortmann, 2001). One shot games do not allow for learning. By repeating the same identical period or by playing a dynamic game (i.e. the situation of the game in period *n* depends on the decisions taken during the *n*-1 previous periods), the players can accumulate information and thus increase their experience.
- The second attribute "ILLUSTRATION" –is the level of information provided in the instructions. Its influence is examined in the literature via the "framing effect" mentioned previously (Tversky and Kahnemann, 1981; Kuhberger and Tanner, 2010).
- **3.** The third attribute **"COMMUNICATION"** considers the fact that when they are assembled in groups in which individual decisions affect the collective results, players can share information.

This exchange can take place in person or using IT media (audio, video or messenger) and causes a change in the way the players understand the game, irrespective of whether it is improved or disturbed (Cardenas, 2003; Carpenter et al., 2004; Ostrom, 2006). This attribute can be controlled: the minimum level is obtained in the laboratory by isolating the players in compartments (no visual contact) and preventing them from speaking to each other. In these conditions, the only exchange is a signal sent (at the end of each period) by each player summarising the decisions taken by him.

4. Finally, the fourth attribute- "EXPERIENCE" – considers the level of information acquired by the player before he joins the game. Experience influences the player's interpretation of the information provided in the instructions, exchanged with his partners and acquired during the repetitions (Gilboa and Schmeidler, 1997; Harrison and List, 2004).

This analytical framework composed of four dimensions of the context (Figure 1), which we call RICE from the initials of each attribute, can be used to test in the laboratory the causal impact that each dimension has on players' behaviour.

In fact, each attribute of the context contributes to the complexity of the protocol. The experimental methodology allows isolating each of these attributes and assessing their influence on the players' behaviours. Based on the fact that an experimental protocol can have different levels of contextualisation (Wang, 1996; Harrison and List, 2004), the impact of each of these attributes can be tested separately using the same protocol. Comparing the results obtained in this way enables assessing the impact of the four dimensions of context on players' behaviour.

Level of complexity of the protocol

Information accumu	Information	Information exchang	Information	
lated over the series	provided through the	ed among the players	provided by players when	F \
of periods	instructions	COMMUNICATION	joining the game	
REPETITION	ILLUSTRATION		EXPERIENCE	

Figure 1. The RICE analytical framework representing the four dimensions of the context in a protocol

2. Coordination in a cooperative game theory set-up

2.1 Origins of our experiment

According to Colman (2006), "A coordination game is an interactive decision in which two or more players have a common interest in coordinating their actions and their expectations of one another's actions". Following this definition, a coordination game can be something different from the class of competitive games with multiple pure strategy Nash equilibria in which players choose the same or corresponding strategies.

Our experimental protocol is based on a coordination game derived from the degradation¹ of a RPG called "KatAware" (Farolfi and Rowntree, 2007), designed and used within a Companion Modelling (Bousquet et al., 2011) project in South Africa to deal with the allocation of water resources in a catchment area. Based on some regularities observed during the RPG sessions and referring players' behaviour to Cooperative Game Theory (CGT), a Transferable Utility (TU) CGT model (Dinar et al., 2008) was implemented. The resulting 3-player coordination game where players can choose among

¹For more information on this approach, cf. Désolé (2011).

a discrete number of options for payoff allocation in a super additive framework is based on the model by Dinar et al. (2008) and allows verifying whether players behave following the CGT theoretical reference, i.e. on the basis of the common advantage, coordinate their actions and reallocate among them the payoff obtained together in an efficient and equitable way.

The solutions of such a game can be presented in the form of a set of allocations complying with certain conditions. Among these solution sets, the *core* (Gillies, 1953) consists of several elements satisfying both the conditions of coalitional rationality and of efficiency, which guarantee the superadditivity² of the game and satisfy the conditions of rationality (individual and coalitional). If the *core* consists of several allocations, the players are asked to choose the one which seems the most equitable. Several single solution concepts are proposed by CGT to identify a solution from the set proposed: the nucleolus, the kernel, the "T-value" and the Shapley value. The Shapley value (1953) is calculated for each player according to his participation in the grand coalition, evaluated by means of the average of his marginal contributions to each of the coalitions of which he is a member. The unit to be taken into consideration for the calculation, therefore, corresponds to what a player contributes to a coalition by joining it. According to the Shapley value, the amount that player *i* gets given a coalitional game (v,N) is:

$$\Phi_{i}(v) = \sum_{S \subseteq N \setminus \{i\}} \frac{|S|!(n-|S|-1)!}{n!} (v(S \cup \{i\}) - v(S))$$

Where *n* is the total number of players and the sum extends over all subsets *S* and *N* not containing player *i*.All other terms refer to note 2.

2.2 The coordination set-up

Our objective was not to observe the coordination process itself but rather to test hypotheses that relate to the impact of introducing the contextual elements presented in the RICE framework on the decisions taken by players. As one of those contextual dimensions is represented by the information that players accumulate over a game session (Repetition), we needed to repeat easily a certain number of periods in a controlled environment using the same players. For this reason we adopted a 3-player game like Rapoport and Kahan (1976) did, but with a different game structure.

Games in which it is possible to create partial coalitions, such as that of Rapoport and Kahan (1976), require very long time frames. In our protocol, the 3 players automatically play together at the start of each period (i.e. the grand coalition was assumed as a given³). Partial coalitions⁴ are not played and the information concerning them is provided to players for the purpose of mutual information only.

The structure of our cooperative game is presented in table 1.

²Where *N* is a finite set of n players (or "grand coalition") in a game with transferable utility, *S* and *T* are subsets of *N* (or sub-coalitions with $S \cap T = \emptyset$) and v(S) is the payoff obtained by the coalition *S*. A game is superadditive if for all coalitions *S* and *T* included in *N*, $v(S \cup T) \ge v(S) + v(T)$.

³As the game is superadditive, it is a necessary consequence of the hypothesis of player rationality.

⁴ The instructions avoid the term "coalition", which is referred to as an "association" or "group". Cf. the detailed instructions in the annex.

	Player A earns 75,000 ecus
Each member of the group remains independent	Player B earns 100,00 0 ecus
	Player C earns 200,000 ecus
Players A and B work together	The association A-B earns 175,000 ecus
	Player Cearns 200,000 ecus
Players A and C work together	The association A-C earns 275,000 ecus
	Player Bearns 100,000 ecus
Players <i>B</i> and <i>C</i> work together	The association B-C earns 350,000 ecus
	Player Aearns 75,000 ecus
The 3 members of the group work together	The association A-B-C earns 500,000 ecus

Table 1.Structure of the cooperative game

As this CGT TU 3-player game structure is superadditive following the definition provided above, the players have a real incentive to cooperate within the grand coalition.

The terminology of CGT defines the payoffs allocation as a function $x = (x_1, ..., x_n) \in \mathbb{R}^N$. The allocation choices therefore belong to the continuous space \mathbb{R}^N . In order to ensure that our protocol is more playable (in particular to overcome the absence of communication between the players), different choices for the grand coalition's payoff allocations are proposed using cards, among which at each period the players must choose the allocation they prefer. The use of cards represents a discretisation of the choices, breaking the condition of continuity of cooperative games and therefore making our protocol rather a coordinationgame. The agreement among players is underpinned by the unanimity rule: if three players all coordinate together on the same card, the corresponding amounts are allocated; if not, each one receives the gain associated with the players playing as singletons (status quo).

	Card 1	Card 2	Card 3	Card 4	Card 5	Card 6	Card 7
Player A	498,000	117,000	1,000	100,000	166,666	1,000	75,000
Player B	1,000	142,000	1,000	150,000	166,666	498,000	100,000
Player C	1,000	241,000	498,000	250,000	166,666	1,000	200,000

The seven proposals (cards) of payoff allocation are as follows:

Table 2. Cards available to the players: amounts in ecus

The numbering on the cards, randomly chosen, is set to be identical for each treatment of the protocol described in the following section. Card 7 does not propose a distribution of the grand coalition's payoff, instead offering the *status quo* that leaves the players with the payoff that they get remaining singletons. Choosing card 7 means choosing non-cooperation, being it explicit or non-intentional following the failure of the group members to coordinate. Cards 1 to 6 are based on different criteria of the distribution of the 500,000 ecus of the grand coalition⁵. These criteria are explained below.

⁵ Players must coordinate themselves (i.e. choose the same card) to reach a consensus for the allocation of the grand coalition payoff. However, players can be willing to stay in the grand coalition, but they can have different ideas in

• "Selfish" criterion: cards 1, 3 and 6

A player proposes to leave only the minimum (1,000 ecus) to his partners in the grand coalition and to keep the rest, i.e.:

This selfish choice is independent of the role and values received as a singleton. Each player can therefore choose to be selfish, hence the fact that three symmetrical cards are proposed, each one favourable to one of the three players. These cards propose a very unfair allocation of the grand coalition payoff to the advantage of the proponent. It could be interpreted as a provocative choice, not really aiming at preserving the coalition.

• "Strict egalitarian" criterion: card 5

The egalitarian criterion often appears in the results of economic experiments, although it is not provided for by the theory for which the hypotheses are tested. Gamson (1964) proposes an egalitarian (or anti-competitive) theory which hypothesises that, rather than attempting to maximise their utility function, the players within a coalition attempt to minimise conflicts to maintain the social link within the group. In the case of super-additive games, egalitarian sharing of the grand coalition's payoff is the ultimate form of anti-competitive sharing. Card 5 is designed as follows:

Each player in the grand coalition receives exactly the same amount rounding down to the nearest ecu, where $x_i = 166,666 \text{ ecus}^6$. The condition of individual rationality is not respected. Player *C* receives an allocation of $x_c = 166,666$ ecus, less than v(C) = 250,000 ecus. This allocation is not in the *core* and therefore is not expected as an outcome by the CGT.

• "Intermediate egalitarian" criterion:card 2

Gamson (1964) considers that strict egalitarian sharing is excluded from superadditive games in which the initial endowments are unequal, because the potential appearance of conflicts of interest is high. Supposing that coalitions form "along the lines of least resistance" (Gamson, 1964), we can consider that the line of resistance for each player, corresponds to the payoff received when that player operates as a singleton. An intermediate form of the anti-competitive theory may demonstrate the criterion of egalitarian sharing beyond these lines of resistance, i.e. an equal division of the surplus of the grand coalition (in relation to the situation where players are singletons). Card 2 is designed as follows:

The sum of the payoffs received when the players operate as singletons is:

$$v(A) + v(B) + v(C) = 375,000$$
 ecus

terms of payoff allocation. In particular, some cards (i.e. 1,3 and 6) propose an allocation plan that is clearly representative of a willingness to break the grand coalition. The choice of these cards would then show that, despite the superadditivity of the game, a player has either not understood the interest of cooperation or is not prepared to coordinate his efforts with the others.

 $^{^{6}}$ The sum of the shares of the grand coalition distributed to the players is thus equal to 3 * 166,666, i.e. 499,998 ecus. We consider that the difference of 2 ecus between the sum of the shares actually allocated by card 5 and the payoff of the grand coalition is negligible: the difference of 2 ecus represents 0.0004% of the 500,000 ecus of the grand coalition.

The surplus of the grand coalition in relation to the singletons situation is the difference between the two situations:

The surplus is shared equally between the 3 players, i.e. Δ / 3 = 125,000 / 3 = 41,666 ecus each, in addition to the payoff obtained in the singletons situation.

Player A will receive v(A) + 41,666 = 116,666 ecus rounded up to x_A = 117,000 ecus.

Player B will receive v(B) + 41,666 = 141,666 ecus rounded up to x_B = 142,000 ecus.

Player *C* will receive v(C) + 41,666 = 241,666 ecus rounded down to x_c = 241,000 ecus.

The allocation of card 2 complies with the conditions of efficiency, individual rationality and coalitional rationality; it is located in the *core* and therefore is an expected outcome in terms of CGT.

• "Equitable and efficient" criterion according to the Shapley value: card 4

We have already seen that the Shapley value is calculated for each player on the basis of his participation in the cooperation, evaluated by means of the average of his marginal contributions to each of the coalitions of which he is a member.

The allocation $x = (x_A, x_B, x_C)$ of card 4, calculated according to the Shapley criterion $\phi = (\phi_A, \phi_B, \phi_C)$, takes into account all the levels of cooperation (singletons, partial coalitions and grand coalition) and complies with the conditions of efficiency, individual rationality and coalitional rationality; it is located in the *core* and therefore is an expected outcome in terms of CGT.

3. The experimental protocol

3.1 Treatments

The protocol is designed on the basis of the coordination gamepresented in the previous section to a) capture the different dimensions of the context defined in the RICE analytical framework: repetition (facilitating the learning process), illustration, communication, and experience, and b) measure the causal impact of each dimension (explanatory variables) on participants' behaviour.

As the causal stimulus of our experiment is represented by the introduction into an abstract⁷ protocol (control, coded here as 0.0) of the four dimensions of context presented in section 1, we built different treatments containing these dimensions and contrasted the outcomes of the treatment groups with those of the control group.

The possible emergence of a *learning effect* due to the accumulation of information during the session was tested through the *repetition* within each treatment (including the control) of 15 periods, each player retaining his role in the same group of 3 subjects during the whole treatment ("partner" mode). The effect of *illustration* was tested by contrasting the outcome of the control (0.0) first with the results of treatments illustrated in an elementary manner by means of a single phrase ("the experiment focuses on a problem of water resource management") (treatment 0.1), then by a narrative explaining the sources of heterogeneity of the individual gains in the water sector (different productivity of water – treatment 0.2), and finally by a narrative explaining the sources of heterogeneity in the industrial sector (different seniority of employees – treatment 0.3). The effect of

⁷ We call the control "abstract" in contrast with the "contextualized" treatments, which are characterized by the introduction of various attributes of context.

communication was tested allowing the players to exchange information⁸ before periods 1, 6 and 11 in a non-illustrated treatment (1.0) and in the most illustrated treatment in the water sector (1.2).

Two important aspects are worthwhile noticing at this stage: first, as we deal with university students and not with expert subjects, the "E" dimension of our RICE framework is not explored so far, and consequently we can rename RIC the framework used for this specific experiment. Second, the R dimension is present in all treatments including the control starting from period 2. The treatment 0.0 is therefore the control only at period 1, as only at this stage all the RIC dimensions are absent.

3.2Hypotheses

Following CGT, players facing a superadditive situation coordinate their choices, stay in the grand coalition and choose to share the resulting payoff following the Shapley value (most efficient and equitable solution). In the control (0.0) at period 1, without illustration, communication and prior to any repetition, we might expect a certain consensus around this theoretical solution. Our experiment then tests first whether theoretical expectations are confirmed in the control (0.0 at period 1).

All the following hypotheses to be tested in the experiment concern the progressive introduction of context dimensions (RIC) through treatments.

Repetition should allow a *learning effect* in players (Cooper and Kagel, 2003). This effect changes player's behaviour in the direction of an improved respect of the theoretical expectations (Andreoni, 1995; Laury et al., 1999)(**H0**). Introducing an illustration is likely to cause greater uncertainty concerning the choices made (Babcock et al., 1995; Binmore, 1999). Adding this contextual dimension in the experiment creates *noise*that increases the dispersion of players' responses in selecting a payoff allocation mechanism if compared to the control (**H1**). This *noise* increases as the illustration of context becomes richer (**H2**). The *noise* does not depend on the specific content of the illustration (**H3**). As communication should facilitate coordination (Brosig et al., 2003; Bochet et al., 2006), we can make the hypothesis that communication does operate a contrary effect to refocus the choices on a preferred payoff allocation configuration (**H4**).

3.3 Experimental conditions

The participants in the sessions were chosen from a database of subjects containing the third-year students enrolled at the universities in Montpellier and provided by the Laboratory of Experimental Economics in Montpellier (L.E.E.M.). We consider that the samples are homogenous and consist of subjects with the same experience concerning the context studied. Each treatment brings together 12 groups of 3 players whose role within the group remains the same throughout the game session ("partner" procedure). These players only participate in one treatment to ensure the independence of the results ("between subjects" procedure).

At the start of the session, the instructions (cf. Annex) are provided in printed format. The participants read the instructions individually then the experimenter reads them aloud. The players are given a comprehension test on the computer to check that they have understood the instructions. The rest of the session is managed by computer.

The session takes place over 15 identical periods. For each period, the individual choices and the cards which establish the payoff allocation to the players are displayed on the screen for the members of the group (but not the results of the other groups) before starting the subsequent

⁸ Information was exchanged by players through "constrained communication", inspired by the "Numerical Cheap Talk (NCT) communication" used by Bochet et al. (2006). Details on the adoption of this method can be found in Désolé (2011).

period. The learning process during the successive sessions therefore relates to the preferences of the other players in the group and can be conducive to attempting to coordinate with regard to an allocation (one card, one behaviour in accordance with the expectations of the other members who form these small societies of 3 players).

The players are paid at the end of the 15 periods in accordance with their performance, which depends on the choices they will have made throughout the game, at a rate of ≤ 0.6 for 100,000 ecus. For instance, the sum of ≤ 3 is distributed to each group of players every time the group succeeds in agreeing on the same means of sharing the 500,000 ecus of the grand coalition.

3.4 Theoretical predictions

Following CGT predictions, in the proposed superadditive set-up participants should coordinate around cards representing payoff allocations situated in the *core*, such as card 4 (Shapley) preferentially, or even card 2 (intermediate egalitarian).

Using the Gately index of "propensity to disrupt the grand coalition"⁹, we can predict the interest of the three players in a grand coalition to coordinate around cards 2 and 4 (Tab. 3):

	Card 2	Card 4
Player A	0.79	2
Player B	1.98	1.5
Player C	2.05	1.5

Table 3.Gately indices (propensity of players to disrupt the coalition) for the allocations of the grand coalition's payoff in the *core*

In table 3, we observe that the propensity of player A ($G_A = 2$) to disrupt the coalition of card 4 (Shapley allocation) is higher than for players B and C ($G_B = G_C = 1.5$). Players B and C have a high propensity (G_B and $G_C \approx 2$) to disrupt the coalition of card 2 ("intermediate egalitarian" allocation) when that of player A is lower ($G_A = 0.79$). Players B and C have similar preferences, oriented towards card 4 rather than card 2. The analysis of the Gately indices (in which the payoffs v(N-{i}) of the

$$G_{i} = \frac{\sum_{j \neq i} x_{j} - v(N - \{i\})}{x_{i} - v(\{i\})}$$

⁹Gately (1974) introduced the concept of "propensity to disrupt the grand coalition". He defined this concept as the relationship between: (a) what the other players lose if player *i*chooses not to cooperate and (b) what player *i* loses in this case (i.e. if he refuses to cooperate). If *N* is the grand coalition comprising *n* players, for all $i \in N$, the propensity G_i to disrupt the coalition $(x_1, ..., x_n)$ can be expressed as follows:

where x_i is the allocation received by player i once the payoff v(N) of the grand coalition has been shared, $v(\{i\})$ is the payoff received by i when it plays as a singleton and $v(N-\{i\})$ is the payoff of the coalition comprising all the players except i.

partial coalitions are taken into account) therefore highlights the symmetry of preferences of *B* and *C*, reflecting the fact that the payoff v(BC) is more advantageous for these two players than the payoffs of the partial coalitions that each one forms separately with player A - v(AB) and v(AC) respectively.

While player *A* has considerable "power of disruption" in the stability of the choice of card 4, he has less power in the tacit bargaining with players *B* and *C* within the grand coalition due to his weak position in the partial coalitions. Although he prefers card 2, card 4 is not the worst solution for him and this should therefore be the resulting point of coordination, confirming the prediction of cooperative game theory which considers the Shapley allocation (represented by card 4) as the players' preferred allocation.

4. Results

4.1 Measure of the dispersion of players' choices between treatments

The factual effects of the introduction of a variable (attribute of the context) on players' behaviour are measured through the difference between the outcomes (i.e. players' choices of cards proposing different payoff allocations) of the treatment characterized by that variable and the outcomes of the control¹⁰.

The experiment outcomes from the control group (0.0) at period 1 are contrasted first with the results from treatments (0.1, 0.2, 0.3), which introduce progressively *illustration*, and then with the results from treatments (1.0, 1.2), which allow *communication*. The *repetition* of periods in all treatments tests the emergence of a *learning effect*, and therefore outcomes between and within all treatments at periods 1 and 15 are contrasted as well.

As we are examining the distribution of choices and not the individual components thereof, in order to assess the difference between the results of the various treatments (e.g. the causal effects of the studied variables), we favour a Chi-squared test.

If m_1 and m_0 are two distributions of the *n* individuals of a population split into *k* classes (*j*=1,2,..*k*)in two distinct circumstances (treatments), then:

$$\chi^{2}(\mathbf{k}-1) = \sum_{j=1}^{k} \frac{(m_{1j} - m_{0j})^{2}}{m_{0j}}$$

Where m_{ij} represents the number of individuals in class j in the i treatment (i=0, 1). $DF=(k-1)^{11}$.

In our analysis, we only examine the distribution of the choices without distinguishing either the roles allocated to the players (A, B or C) or the failure or success in coordinating within the grand coalitions. In other terms, we observe the players' proposals to coordinate (or not) in order to allocate the grand coalition payoff, we do not analyse the actual coordination that follows. Twelve groups of 3 players took part in each treatment, i.e. 36 players (n). For each treatment, we will therefore analyse the distribution of the 36 choices expressed between the 4 classes of cards (k): class 1 (cards 7,1,3,6) corresponding to the refusal to cooperate; class 2 (card 5) corresponding to the allocation strictly egalitarian of the payoff of the grand coalition; class 3 (card 2) corresponding to the

¹⁰The session conducted with the control group (i.e. in the absence of context attributes) in the text below is called control.

¹¹We grouped cards 1, 3, 6 and 7 together because they are based on a similar criterion (refusal to cooperate). Card 7 means in fact a direct and explicit refusal to stay in the grand coalition, while cards 1,3, and 6 propose a *outrageously*unfair allocation of the grand coalition payoff to the advantage of the proponent, and have therefore no chance to be accepted by the two other players. We therefore have K = 4 and thus K-1 = 3 degrees of freedom.

intermediate egalitarian allocation of the payoff of the grand coalition (in the *core*); and class 4 (card 4) corresponding to the allocation of the payoff of the grand coalition that follows the Shapley value (in the *core*).

It is worthwhile noticing that the choice to coordinatebetween players in this protocol takes place at three progressive levels: i) staying in the grand coalition allocating the consequent payoff in a "reasonable", i.e. not selfish, way (cards 2, 4 and 5); ii) choosing a payoff allocation within the *core* (cards 2 and 4); iii) choosing the Shapley value as a payoff allocation criterion (card 4).

4.2 Respect of theoretical predictions in the controlgroup

Before moving into the test of our hypotheses about the influence of context on players' behaviour, we needed to verify whether the theoretical predictions are respected within the control group (0.0).

We therefore checked the distribution of players' choices at period 1, which is the only situation where all dimensions of context following the RICE framework are absent.

Period	Cards Treat- ment	7,1,3,6	5	2	4
1	0.0	6	22	4	4

 Table 4.Control group's choices at period 1¹²

Table 4 shows that 30 players out of 36 choose to coordinate and stay in the grand coalition, but 22, the large majority, choose card 5 (strictly egalitarian) rather than the payoff allocation solutions which reside in the *core* according to CGT.

This result can be interpreted as a *Schelling effect* (Schelling, 1960). The *Schelling effect* consists of the emergence of a *focal point* for the coordinating players far from the theoretical predictions. "A combination of actions may function as a *focal point* if it stands out from others because of some property of prominence or salience that it possesses, even if its distinguishing feature is a psychological or cultural attribute that is filtered out of its abstract representation in orthodox game theory" (Colman, 2006).

In the absence of communication and at the beginning of the sessions, a certain consensus around a sharing mechanism (*focal point*) other than the Shapley value, or even other than a mechanism residing in the *core*, characterised the choices of players.

This important result puts our protocol into a situation where the control does not produce a full convergence of results towards the theoretical predictions (card 4 or even card 2 in the *core*). Subsequently, a dispersion of results in the treatments due to the *noise* provoked by the introduction of contextual dimensions does not necessarily mean an increased distance from the theoretical predictions. It is then possible to find out that the introduction of contextual dimensions influences players' behaviour positively, increasing the convergence of choices towards the theoretical equilibrium rather than decreasing it.

¹² The cards are presented in tables 4 to 11 following the same order as in section 2.2, where they are explained.

4.3 Repetition of periods and *learning effect*

At period 15, players of the control group (0.0) shift their choices from card 5 to cards 4 and 2, while the refusals to cooperate are divided by two (Table 5). This result indicates the existence of a *learning effect* between period 1 and period 15 (confirmed by the high Chi-squared calculated between the two distributions). That learning, in the absence of all other contextual dimensions, increases players' willingness to coordinate their choices around an acceptable allocation of the grand coalition payoff (sum of choices of cards 5, 2, and 4) and facilitates the reach of the theoretical predictions (sum of choices of cards 2 and 4).

We checked whether the same *learning effect* is observable in the other treatments as well. The Chisquared test shows clearly that the probability that the two distributions of choices at periods 1 and 15 are different is 100% or close in all treatments, confirming the existence of a *learning effect* also when illustration and communication are introduced. **This result would confirm H0**¹³.

Period	Cards Treat- ments	7,1,3,6	5	2	4	χ²	Prob
1	0.0	6	22	4	4	51 20	1 000
15	0.0	3	10	6	17	51.25	1.000
1	0.1	8	22	1	5	76.68	1 000
15	0.1	8	9	9	10		1.000
1	0.2	3	23	6	4	10 / 2	0 000
15	0.2	3	12	11	10	10.42	0.999
1	0.3	3	23	7	3	14 50	0.007
15	0.3	1	15	15	5	14.55	0.337
1	1.0	4	19	1	12	27.90	1 000
15	1.0	6	6	6	18	37.89	1.000
1	1.2	7	16	7	6	25.15	1 000
15	1.2	1	5	17	13	55.15	1.000

Table 5. χ^2 (3) test and players' choices at periods 1 and 15 of alltreatments

The fact that the university students share an industrial culture rather than the culture of the irrigators, might be at the origin of the very marked *Schelling effect* at period 15 in treatment 0.3, with choices focusing on the "egalitarian" proposals (i.e. card 5 and card 2, together representing 83% of the choices) to the detriment of card 4 (Shapley).

¹³ Notice that in an experiment where 0.0 at period 1 produces results in line with the theoretical predictions, H0 is expected to be rejected.

4.4 Illustration

To evaluate the "pure" illustration effect (without learning, without communication), we compared the distributions at period 1 between the control (0.0) and the illustrated treatments without communication (0.1, 0.2 and 0.3).

Period	Cards Treat- ments	7,1,3,6	5	2	4	χ²	Prob
1	0.0	6	22	4	4	3.16	0.633
	0.1	8	22	1	5	5.10	
1	0.0	6	22	4	4	254	0 532
	0.2	3	23	6	4	2.34	0.332
1	0.0	6	22	4	4	4.04	0 7/3
	0.3	3	23	7	3	4.04	0.743

Table 6. χ^2 (3) test at period 1 between treatments(0.0) and (0.1), (0.2), (0.3)

The low Chi-squared statistics (Tab. 6) show that there is very little difference in the distribution of players' choices between the control and the illustrated treatments. The "pure" illustration effect is therefore very weak. The players focus their decisions on card 5 irrespectively of the illustration provided (or not) in the instructions. "Pure" illustration does not have a sufficiently strong effect on the decisions to correct the *Schelling effect* observed in the abstract treatment (0.0) in favour of card 5 (egalitarian). **This result would reject H1**¹⁴ as no *noise* provoked by the illustration seems to influence the outcomesof treatments 0.1, 0.2 and 0.3 at period 1.

Period	Cards Treat- ments	7,1,3,6	5	2	4	χ²	Prob
1	0.1	8	22	1	5	70 27	1.000
•	0.2	3	23	6	4	20.57	
1	0.2	3	23	6	4	0.41	0.063
	0.3	3	23	7	3	0.41	0.005

Table 7.χ²(3) tes	t at period 1 betweer	treatments(0.1)-(0.2)
-------------------	-----------------------	-----------------------

and between treatments (0.2)-(0.3)

Comparing the results of treatments (0.1) and (0.2) allows **confirming H2**, as the significant Chisquared between the outcomes demonstrates that *noise* increases as the illustration of the same context becomes richer. On the other side, there is no significant difference between the outcomes of treatments (0.2) and (0.3), showing that the increase in *noise* does not depend on the specific

¹⁴ Notice that in an experiment where 0.0 at period 1 produces results in line with the theoretical predictions, the confirmation of H1 would correspond to a dispersion of treatments' outcomes far from the allocations in the *core*.

content of the illustration, which moves in the experiment from a water-related problem (0.2) into an industry-related problem (0.3). This would **confirm H3**.

The result is different when the illustration effect is combined with the *learning effect* resulting from repetition of periods. Showing this combined effect is possible through the comparison of the distribution of choices in period 15 for (0.0) and (0.1).

Period	Cards Treat- ments	7,1,3,6	5	2	4	χ²	Prob
15	0.0	3	10	6	17	12 01	0.004
15	0.1	8	9	9	10	12.01	0.554

Table 8. χ^2 (3) test at period 15 between treatments(0.0)-(0.1)

At the end of the session even a very light introduction of context in the protocol (the phrase "the experiment focuses on a problem of water resource management") creates a *noise* and a consequent significant change in the behaviour of players captured by a relatively high Chi-squared at period 15.

4.5 Communication

To measure the "pure" communication effect (without illustration, without learning), we compare the distribution of choices at period 1 of the control with the treatment non illustrated but allowing communication (1.0) (Tab. 9).

Period	Cards Treat- ments	7,1,3,6	5	2	4	χ²	Prob
1	0.0	6	22	4	4	10.22	0 000
•	1.0	4	19	1	12	19.52	0.333

Table 9. χ^{2} (3) test atperiod 1 between treatments(0.0)-(1.0)

If we allow the subjects to exchange ideas, we observe a significant communication effect as early as at period 1. The choice of an egalitarian allocation (card 5) remains high, but there is an increased preference for the Shapley allocation (card 4) from the very first round. The refusals to cooperate also reduce from 6 to 4.

Period	Cards Treat- ments	7,1,3,6	5	2	4	χ²	Prob
15	0.0	3	10	6	17	4.65	0 801
15	1.0	6	6	6	18	4.05	0.001

Table 10. χ^2 (3) test atperiod 15 between treatments (0.0)-(1.0)

At period 15 of a non illustrated treatment, communication seems to have exhausted its effect (Tab. 10). The distribution of players' choices does not differ whether it is in the presence of communication or not. It is like if communication just speeds-up the reach of the theoretical prediction.

Let us now consider the effect of introducing communication in an illustrated treatment. This is possible by comparing treatments(0.2) and (1.2) (Tab. 11).

Period	Cards Treat- ments	7,1,3,6	5	2	4	χ²	Prob
1	0.2	3	23	6	4	8.63	0.965
•	1.2	7	16	7	6		
15	0.2	3	12	11	10	9.58	0 977
	1.2	1	5	17	13		0.377

Table 11. χ^2 (3) test atperiods 1 and 15 between treatments (0.2)-(1.2)

We observe that distinct distributions of players' choices exist both at periods 1 and 15, showing the effect of communication. The probability that distributions follow the same law is rejected at the 5% level.

The results presented in Tab. 11 show that communication has an effect when introduced in illustrated treatments. But does this effect represent a *noise* as defined by Babcock et al. (1995), i.e. a disruption of the participants' behaviour observed in the absence of all context attributes? Figures presented in table 9 answer positively for period 1, while if we compare the distribution of players' choices at period 15 in treatment (1.2) and in the control we get a Chi-squared of 24.2, which is higher than the Chi-squared (7.4) resulting from the comparison at period 15 between treatment (0.2) and the control. Communication therefore amplifies and does not reduce the *noise* produced by the pure illustration on the control. **This conclusion would reject H4**¹⁵.

4.6Combined effects of contextual attributes

In an experiment where the control at period 1 produces results in line with the theoretical predictions, the study of the combined effects of contextual attributes is relatively straightforward, as repetition does not need to produce a *learning effect*to make the players converge towards theoretical predictions, while the *noise* increases the distance between the treatments' results and the theoretical predictions.

In our experiment the control did not produce results in line with the theoretical predictions, and as a consequence the *learning effect* produced by repetition was important in changing the behaviour of players in the direction of the theoretical predictions. At the same time, the *noise* could have both effects of increasing or reducing the distance between the treatments' outcomes and the theoretical predictions.

¹⁵ Notice that in an experiment where (0.0) at period 1 produces results in line with the theoretical predictions, the rejection of H4 would correspond to a dispersion of results in (1.0) and (1.2) if compared to the theoretical predictions. In the presented case, vice versa, the rejection of H4 corresponds to a convergence of results in (1.0) and (1.2) towards the theoretical predictions.

The combined effect of the contextual attributes was therefore more articulated. At period 1, for instance, the combination of illustration and communication (Tab. 11, first two rows) produced a *noise* and at the same time allowed a slight convergence of results towards the theoretical predictions (choice of cards 4 and 2 in the *core*).

When repetition was introduced (at period 15), the *learning effect* was clear in all treatments (Tab. 5). However, the combination of the *learning effect* with the effects of illustration (Tab. 8) and communication (Tab. 10) showed that communication had a lower influence than illustration in terms of *noise* creation (low Chi-squared) and a higher influence in terms of convergence towards theoretical predictions (less refusals to stay in the grand coalition, more choices of cards in the *core*).

These results were confirmed by the outcomes presented in Tab. 5, where the strongest *learning effect* (highest Chi-squared) was observed either in the abstract treatment (0.0) or when minimum illustration was provided (0.1), while the weakest *learning effect* (lowest Chi-squared) corresponded to the illustrated treatments without communication (0.2 and 0.3). Conversely, the best convergence towards theoretical predictions (highest choice of cards in the core) corresponded to the treatments where communication was allowed (1.0 and especially 1.2, where the combined effect of repetition, illustration and communication coexists), while the worst convergence was in presence of illustration alone (0.1, 0.2, and 0.3).

Our experiment, while confirming the production of *noise* (Babcock et al., 1995) through the introduction of contextual attributes in treatments, seems therefore to corroborate the hypotheses made by Faravelli (2007) and Loewenstein (1999) that context provides players with indicators enabling them to behave in accordance with a common representation of this context. According to Cooper and Kagel (2003), when learning processes are allowed, a "weak" contextual effect would also allow behaviours to converge more quickly towards theoretical predictions. In our experiment, however, illustration (which often in EE is the only context considered) proved to have lower effect than communication in terms of accelerating learning processes.

5. Conclusion

The objective of this paper was twofold. We aimed first to provide a framework for the analysis of contextual components in an economic experiment. Then, based on the mentioned framework, we wanted to find out quantitatively and objectively whether and how the context has a causal impact on players' behaviours.

To do so, an experimental protocol based on a 3-player CGT TU game modified into a coordination game to improve its playabilitywas constructed. The game is about the grand coalition payoff allocation among players in two different situations: one about water and another about an industrial context. Several treatments referring to the first three contextual attributes of the four (Repetition, Illustration, Communication, and Experience) included in the proposed analytical framework were designed and implemented in the laboratory to isolate the causal effect of each attribute on players' behaviour.

We wanted to test the following hypotheses: Repetition determines a *learning effect* (H0); Introducing an illustration increases the *noise* in terms of experiments' outcomes, i.e. the dispersion of these outcomes in the treatments w.r.t. the control (H1); This *noise* grows as the illustration of the same context becomes richer (H2); The increase in *noise* does not depend on the specific content of the illustration (H3); Communication does operate a contrary effect to refocus the choices on a preferred payoff allocation configuration (H4).

Results from the experiment confirmed H0, rejected H1, showing that illustration alone does not produce a significant *noise* in terms of treatment's outcomes. H2 and H3 were confirmed, as more illustration was positively correlated with higher dispersion of outcomes w.r.t. the control, while the

specific topic of the illustration did not seem to influence that dispersion. H4 was also rejected as communication did not seem to be able to reduce the *noise* produced by illustration, but rather increased it.

In our experiment the control did not produce results in line with the theoretical expectations(Shapley value, or even a payoff allocation residing in the *core* of the efficient and rational allocations), but rather a convergence to a *focal point* (*Schelling effect*) represented by the coordination towards an egalitarian allocation of the grand coalition payoff (card 5). This fact allowed interpreting the results of the treatments from two different perspectives, otherwise merged into the same criterion: 1) the production of *noise* (i.e. the increased dispersion of results in the treatments w.r.t. those of the control), and 2) the convergence of outcomes towards theoretical predictions.

In the absence of repetition, the combination of illustration and communication produced *noise* and allowed at the same time a slight convergence of results towards the theoretical predictions. When repetition was introduced, the *learning effect* was clear in all treatments. However, communication had a lower influence than illustration in terms of *noise* creation and a higher influence in terms of convergence of outcomes towards theoretical predictions.

These results contribute shedding some light, although partially, on the general problem of the quantity and quality of context that are "good" for both experimental protocols and RPGs in order to reach the acceptable compromise between internal and external validity in both disciplines.

It is clear that the results of the present work cannot be generalized, but they represent an initial step in the direction of the interpretation of the causal impacts that selected attributes of the so-called context have on players' behaviour, both in EE and in RPGs.

Among the substantial caveats still remaining and that should be approached in future works in this field, it can be mentioned the relative limited number of observations in the laboratory, which might affect negatively the statistical representativeness of the results obtained. This is also due to the relatively complex and long protocol (repetition of 15 periods for every treatment), which could be reduced to a "one shot" experiment in the future. The number of cards from which to choose the allocation of payoff could also be limited in order to speed and facilitate the interpretation of results.

It is important to notice, finally, that the RICE framework was only partially implemented so far, as the experiments were conducted in the laboratory with university students and not in the field with expert subjects. The "E" for Experience will therefore be approached only when field experiments involving real stakeholders (e.g. farmers for the water problem) will take place.

Acknowledgments

The authorsgratefully acknowledge the financial support from the SAFe Water ARISE project, CIRAD, and the University of Montpellier I.Thank you to the Laboratory of Experimental Economics in Montpellier (LEEM-LAMETA, Montpellier France) where the sessions were conducted in 2009 and 2010, and particularly toDimitri Dubois who provided technical support in preparation of, and during, the sessions, and computerized the protocol in JAVA® language.We wish to thank our colleagues Olivier Barreteau,François Bousquet, Ariel Dinar, Nils Ferrand, Annie Hofstetter, FioravantePatrone, Juliette Rouchier, Sophie Thoyer, and MabelTidballfor theirprecious comments and suggestions provided at different steps of our work.

References

Abrami G., Ferrand N., Morardet S., Murgue C., Popova A., De Fooij H., Farolfi S., and Du Toit D., Aquae-Gaudi W. (2012) Wat-A-Game, a toolkit for building role-playing games about integrated water management, in: R. Seppelt R, Voinov A.A., Lange S., and Bankamp D. (Eds.) Proceedings of the 6th International Congress on Environmental Modelling and Software (iEMSs), Leipzig, Germany, 1-5 July.

Andreoni J. (1995) Cooperation in Public-Goods Experiments: Kindness or Confusion?, The American Economic Review, vol. 85 (4), p. 891-904.

Babcock L., Loewenstein G., Issacharoff S., andCamerer C. (1995) Biased judgments of fairness in bargaining. American Economic Review, vol. 85, p. 1337-1343.

Barreteau O., Bousquet F., Etienne M., Souchère V., and d'Aquino P. (2011) Companion Modelling : A method for adaptive and participatory research, in: Etienne M. (Ed.) Companion Modelling: A participatoy approach to support sustainable development, Quae, Versailles, p. 21-44.

Binmore K. (1999) Why experiment in Economics? The Economic Journal, vol. 109 (February), p. F16-F24.

Bochet O., Page T., and Putterman L (2006) Communication and Punishment in Voluntary Contribution Experiments', Journal of Economic Behavior and Organization, vol. 60(1), p. 11-26.

Bousquet F., Etienne M., and d'Aquino P. (2011) Introduction, in : Etienne M. (Ed.) Companion Modelling: A participatoy approach to support sustainable development, Quae, Versailles, p.11-20.

Brosig J., Ockenfels A., Weimann J. (2003) The effect of communication media on cooperation, German Economic Review, vol. 4(2), p. 217-241.

Campbell D.T. (1957) Factors relevant to the validity of experiments in social settings, Psycological bulletin, vol. 54, p. 297-312.

Cardenas J.C. (2003) Real Wealth and Experimental Cooperation: experiments in the field lab. Journal of Development Economics, vol. 70(2), p. 263-289.

Carpenter J., Burks S., and Verhoogen E. (2004) Comparing Students to Workers: The Effects of Social Framing on Behavior in Distribution Games.IZA Discussion Paper N. 1341.

Colman A. (2006) Thomas C. Schelling's psychological decision theory: Introduction to a special issue, Journal of Economic Psychology, vol. 27, p. 603-608.

Cooper R., DeJong D.V., Forsythe R., and Ross T.W. (1989) Communication in the Battle of the Sexes Game: Some Experimental Results, The RAND Journal of Economics, vol. 20(4), p. 568-587.

Cooper D.J., Kagel J.H. (2003) The impact of meaningful context on strategic play in signalling games. Journal of Economic Behavior and Organization, vol. 50, p. 311-337.

Désolé M. (2011) Théorie des Jeux, jeux de rôles et expérimentations pour la gestion de l'eau : étude expérimentale des effets du contexte sur les résultats d'un jeu de rôles. PhD dissertation, Montpellier SupAgro, 262 pages.

Désolé M., Farolfi S., Patrone F., and Rio P. (2009) From Experience to Experiments in South African Water Management: Defining the Framework, Lameta Working Paper 2009-24, Lameta Montpellier, 14 pages.

Dinar A., Farolfi S., Patrone F., and Rowntree K. (2008) To negotiate or to game theorize: evaluating water allocation mechanisms in the Kat Basin, South Africa, in Game Theory and Policy Making in Natural Resources and the Environmet, ed. Dinar, Albiac and Sanchez-Soriano, Routledge.

Druckman J., Green D.P., Kuklinski J.H. and Lupia A. (2012) Experimentation in political science, in: Druckman J., Green D.P., Kuklinski J.H. and Lupia A. (Eds.) Cambridge Handbook of Experimental Political Science, p. 1-18.

Faravelli M (2007) How context matters: A survey based experiment on distributive justice. Journal of Public Economics, vol. 91, p. 1399–1422.

Farolfi S., Rowntree K. (2007) Accompanying Local Stakeholders in Negotiation Processes related to Water Allocation through Simulation Models and Role-Playing Games: an Experience from South Africa, EMPOWERS, issue2, January 2007.

Ferrand N., Farolfi S., Abrami G., anddu Toit D. (2009) Wat-A-Game: Sharing water and policies in your own basin, 40th Annual Conference International Simulation and Gaming Association, 29 June to 3 July 2009, Singapore.

Gamson W.A. (1964) Experimental Studies of Coalition Formation, Advances in Experimental Social Psychology, vol. 1, p. 81-110.

Gately D. (1974) Sharing the Gains from Regional Cooperation: A Game Theoretic Application to Planning Investment, Electric Power. International Economic Review, vol. 15(1), p. 195-208.

Gilboa I., Schmeidler D. (1997) Act similarity in case-based decision theory, Economic Theory, vol. 9, p. 47-61.

Gillies, D.B. (1953). Some theorems on n-person games, PhD dissertation, Princeton University Press.

Harrison G.W., List J.A. (2004) Field Experiments, Journal of Economic Literature, vol. 42(4).

Hertwig R., Ortmann A. (2001) Experimental practices in economics: A methodological challenge for psychologists?, Behavioral and Brain Sciences, vol. 24, p. 383-451.

Kuhberger A., Tanner C. (2010) Risky Choice Framing: Task Versions and a Comparison of Prospect Theory and Fuzzy-Trace Theory, Journal of behavioral Decision Making, vol. 23, p. 314-329.

Laury S. K., Walker J. M., Williams A.W. (1999) TheVoluntary Provision of a Pure Public Good with Diminishing Marginal Returns, PublicChoice, vol. 99 (1), p. 139-160.

Le Page C., Abrami G., Barreteau O., Becu N., Bommel P., Botta A., Dray A., Monteil C., and Souchère V. (2011) Models for sharing representations, in: Etienne M. (Ed.) Companion Modelling: A participatoy approach to support sustainable development, Quae, Versailles, p. 69-96.

Loewenstein G. (1999) Experimental Economics from the Vantage point of Behavioral Economics. The Economic Journal, vol. 109 (February), p. 25-34.

Loomes G. (1999) Some lessons from past experiments and some challenges for the future, the Economic Journal, vol. 109 (February), p. 35-45.

Mc Dermott R. (2012) Internal and external validity, in: Druckman J., Green D.P., Kuklinski J.H. and Lupia A. (Eds.) Cambridge Handbook of Experimental Political Science, p. 42-69.

Ostrom E. (2006). The value-added of laboratory experiments for the study of institutions and common pool resources. Journal of Economic Behavior & Organization, vol. 61, p.149-63.

Pillutla M.M., Chen X-P. (1999) Social Norms and Cooperation in Social Dilemmas: The Effects of Context and Feedback. Organizational Behavior and Human Decision Processes, vol.48(2), p. 81-103.

Rapoport A., Kahan J.P. (1976) When three is not always two against one: coalitions in experimental three-person cooperative games, Journal of Experimental Social Psychology, vol. 12, p. 253-273.

Roth A. E., Erev I. (1995) Learning in extensive-form games: Experimental data and simple dynamic models in the intermediate term, Games and Economic Behavior, vol. 8(1), p. 164-212.

Rouchier J. (2007) Data Gathering to Build and Validate Small-Scale Social Models for Simulation. In Rennard, J.P., (Ed), Handbook of Research on Nature Inspired Computing for Economics and Management, Hershey Chapter XV, p. 198-212.

Schelling T. C. (1960) The Strategy of Conflict, Cambridge.

Shadish W.R., Cook T.D., and Campbell D.T. (2002) Experimental and Quasi-Experimental Designs for Generalized Causal Inference.Boston: Houghton Miflin.

Shapley L.S. (1953) The Value of an N-Person Game. In H.W. Kuhn and A.W. Tucker (Ed) Contributions to the Theory of Games.Princeton University Press.

Tversky A., Kahneman D. (1981) The Framing of Decisions and the Psychology of Choice, Sciences, vol. 211, N. 4481, p. 453-458.

Wang X.T. (1996) Framing Effect: Dynamics and Task Domains, Organizational Behavior and Human Decisions Processes, Vol 68(2), November, p. 145-157.

Annex

Instructions to players

Control(0.0)

The experiment you will participate is part of a study on decision-making. The instructions are simple. If you follow them and if you take good decisions, you will earn a non-negligible amount of money. All your answers are anonymous and are collected by a computer. You will indicate your choices through a computer in front of you and your earnings will be displayed on the screen. The amount of money you will earn will be provided to you in cash at the end of the experiment.

1) General framework of the experiment

There are 18 participants in the room. At the beginning of the experiment the central computer will randomly gather you in 6 three-person groups. The setting-up of the groups will remain identical along the experiment. You cannot identify the other members of your three-person group and the two others cannot identify you. The experiment will last 15 identical periods.

Your earnings will depend on your own decisions and at the same time on the decisions the two other members of your group will make. Each decision will allow you to earn *ecus*. All the *ecus* you will earn will be changed in *euros* and you will receive them in cash at the end of the experiment. The *ecus-euros* exchange rate is detailed at the end of these instructions.

2) Situations

At the beginning of the experiment, the central computer will assign a role to each member of the group. Then, one member will play role A, another role B and the last role C. In the instructions below, "player A" will designate the participant to whom role A is allocated, "player B" the person to whom role B is assigned and "player C" the one who assumes role C. Each member will know his role at the beginning of the experiment. Everyone keeps the same role along the experiment.

Within each group, three situations are possible:

- (i) Each member remains independent,
- (ii) One member associates with another while the third member remains independent,
- (iii) The three members are associated.

The earnings of the members or the earnings of the association change following the situation as stated in table 1.

Situation	Earnings
	Player A earns 75 000 ecus
Each member remains independent	Player B earns 100 000 ecus
	Player C earns 200 000 ecus
Players A and P are accepted	The association A-B earns 175 000 ecus
Players A allu B are associated	Player C earns 200 000 ecus
Players A and C are associated	The association A-C earns 275 000 ecus
Players A allu C ale associated	Player C earns 100 000 ecus
Disvers D and C are associated	The association B-C earns 350 000 ecus
Players B and C are associated	Player C earns 75 000 ecus
The three members are associated	The association A-B-C earns 500 000 ecus

Table 1 – The earnings according to the different situations

In the experiment, only situations (i) and (iii) are considered.

3) Decisions

At each period you must choose one card among the 7 presented in table 2. Each card displays the earnings of each member of the group depending on his role (A, B or C).

Note that cards 1 to 6 refer to situation (iii) while card 7 refers to situation (i).

	Card 1	Card 2	Card 3	Card 4	Card 5	Card 6	Card 7
Player A	498 000	117 000	1 000	100 000	166 666	1 000	75 000
Player B	1 000	142 000	1 000	150 000	166 666	498 000	100 000
Player C	1 000	241 000	498 000	250 000	166 666	1 000	200 000
				-			

Table 2 – The seven cards

If the three players choose the same card, they earn the corresponding amounts; otherwise, it is like card 7 was chosen by all.

Example 1:

Player A chooses card 7, player B chooses card 5 and player C chooses card 5. The three members of the group do not choose the same card, therefore card 7 is implemented in the group. Player A earns 75 000 ecus, player B earns 100 000 ecus and player C earns 200 000 ecus.

Example 2:

Player A chooses card 1, player B chooses card 1 and player C chooses card 1. The three members of the group choose card 1, therefore this card is implemented. Player A earns 498 000 ecus while players B and C earn 1 000 ecus.

4) The sequence of a period

All the periods are identical. At the beginning of a period, the decision page is displayed on the screen in front of you. You must choose one card among the seven proposed. When all participants have made their decision, a new page appears in the screen which summarizes the following information:

- The card you have chosen and the card your two partners in the group have chosen,
- The card which is implemented for the group,
- Your earnings for the period.

The bottom part of the screen displays the history of the periods, i.e. the summary of the choices made during the previous periods and the cumulative earnings from the beginning of the session. The history of the periods is available from the decision page by clicking on the button "history".

5) Last details

At the end of period 15, your complete earnings will be displayed on your screen and will be expressed in ecus and in euros, knowing that the exchange rate is 100 000 ecus = 0.6 euros. Then, if you earned 2 500 000 ecus through the experiment, you will leave the room with 15 euros in cash.

Treatment (0.1)

In treatment 0.1, only one sentence is added to the instructions at the beginning of section 2 to distinguish them from the instructions distributed to the players in treatment 0.0. Then, in treatment 0.1section 2 starts as follows:

(...)

2) Situations

The experiment is articulated around a water management issue

(...)

Treatment (0.2)

In treatment 0.2, a story was added to the instructions to distinguish them from the instructions distributed to the players in treatment 0.1. The particular sentence added in treatment 0.1 is maintained. Moreover, players are identified as farmers; every word "player" is substituted by the word "farmer" in tables 1 and 2 and in the two examples.

In treatment 0.2, introduction and sections 1, 4 and 5 remain identical. Only sections 2 and 3 change and become as follows:

(...)

2) Situations

The experiment is articulated around a water management issue

You and the two other members of your group are irrigator farmers, meaning that you can irrigate your crop. Water is essential for your production. Water is naturally provided by rain, but irrigation allows you to provide additional water to your crop in order to improve the development of the plant and then to get better yields. Without additional water your production is weaker and therefore your income is lower.

The available water for irrigation is stored in a dam and you share it with the two other farmers of your group. The quantity of water stored in the dam is not sufficient to allow optimal irrigation for the three cultivated surfaces (your surface and the ones of the two other farmers).

An irrigation scheme is implemented. It allows you to build associations of farmers in order to optimize the use of irrigation water from the dam.

You and the two other farmers in your group get the same cultivated surface but with different yields. At the beginning of the experiment, the central computer will assign a yield (A, B or C) to each member of the group. Then, one member will have the yield A, another the yield B and the last the yield C. In the instructions below, "farmer A" will designate the participant to whom the yield A is allocated, "farmer B" the person to whom the yield B is assigned and "farmer C" the one who assume the yield C. Each member will know whether he is farmer A, B or C at the beginning of the experiment. Everyone keeps the same role along the experiment.

Within each group, three situations are possible:

- (i) Each member remains independent,
- (ii) One member associates with another while the third member remains independent,
- (iii) The three members are associated.

The earnings of the members or the earnings of the association change following the situation as stated in table 1.

Situation	Earnings		
	Farmer A earns 75 000 ecus		
Each member remains independent	<i>Farmer</i> B earns 100 000 ecus		
	Farmer C earns 200 000 ecus		
Formers A and B are associated	The association A-B earns 175 000 ecus		
Furmers A and B are associated	Farmer C earns 200 000 ecus		

	1 I 1000 . I I		
The three members are associated	The association A-B-C earns 500 000 ecus		
Furniers B and C are associated	Farmer C earns 75 000 ecus		
Formers B and C are associated	The association B-C earns 350 000 ecus		
runners A and C are associated	Farmer C earns 100 000 ecus		
Formers A and C are associated	The association A-C earns 275 000 ecus		

Table 1 – The earnings according to the different situations

In the experiment, only situations (i) and (iii) are considered.

3) Decisions

At each period you must choose one card among the 7 presented in table 2. Each card displays the earnings of each member of the group depending on his role (A, B or C).

Note that cards 1 to 6 refer to situation (iii) while card 7 refers to situation (i).

	Card 1	Card 2	Card 3	Card 4	Card 5	Card 6	Card 7
Farmer A	498 000	117 000	1 000	100 000	166 666	1 000	75 000
Farmer B	1 000	142 000	1 000	150 000	166 666	498 000	100 000
Farmer C	1 000	241 000	498 000	250 000	166 666	1 000	200 000
Table 2 – The seven cards							

If the three *farmers* choose the same card, they earn the corresponding amounts; otherwise, it is like card 7 was chosen by all.

Example 1:

Farmer A chooses card 7, **Farmer** B chooses card 5 and **Farmer** C chooses card 5. The three members of the group do not choose the same card, therefore card 7 is implemented in the group. **Farmer** A earns 75 000 ecus, **Farmer** B earns 100 000 ecus and **Farmer** C earns 200 000 ecus.

Example 2:

Farmer A chooses card 1, **Farmer** B chooses card 1 and **Farmer** C chooses card 1. The three members of the group choose card 1, therefore this card is implemented. **Farmer** A earns 498 000 ecus while **Farmer** B and C earn 1 000 ecus.

(...)

Treatment (0.3)

In treatment 0.3, the context of the story was changed from water to industry. Farmers become employees in a firm. The specific water related sentence of treatment 0.1 is removed.

In treatment 0.3, introduction and sections 1, 4 and 5 remain identical. Only sections 2 and 3 change and become as follows:

(...)

2) Situations

You and the two other members of your group are *employees in a firm. Open markets and the intensification* of competition conducted to a deterioration of the economic situation of your firm. Jobs are on the line and the top management decides to delegate responsibilities to the employees in order to improve efficiency in the firm.

A new organization of work is implemented in your firm. It allows you to associate with your colleagues in order to optimize the firm's production.

You and two other *employees of your firm* get the same *machinery*, but with different *productivity, due to workers' different skills and experience*. At the beginning of the experiment, the central computer will assign *productivity* (A, B or C) to each member of the group. Then, one member will get *productivity* A, another *productivity* B and the last *productivity* C. In the instructions below, *"employee A"* will designate the participant to whom *productivity* A is allocated, *"employee B"* the person to whom *productivity* B is assigned

and *"employee C"* the one who assume *productivity* C. Each member will know whether he is*employee A, B or C* at the beginning of the experiment. Everyone keeps the same role along the experiment.

Within each group, three situations are possible:

- (i) Each member remains independent,
- (ii) One member associates with another while the third member remains independent,
- (iii) The three members are associated.

The earnings of the members or the earnings of the association change following the situation as stated in table 1.

Situation	Earnings
	Employee A earns 75 000 ecus
Each member remains independent	Employee B earns 100 000 ecus
	Employee C earns 200 000 ecus
Frankauses A and D are accepted	The association A-B earns 175 000 ecus
Employees A and B are associated	Employee C earns 200 000 ecus
Employees A and C are accepted	The association A-C earns 275 000 ecus
Employees A and C are associated	Employee C earns 100 000 ecus
Employees P and C are associated	The association B-C earns 350 000 ecus
Employees B and C are associated	Employee C earns 75 000 ecus
The three members are associated	The association A-B-C earns 500 000 ecus

Table 1 – The earnings according to the different situations

In the experiment, only situations (i) and (iii) are considered.

3) Decisions

At each period you have to choose one card among the 7 presented in table 2. Each card displays the earnings of each member of the group depending on his role (A, B or C).

observe that cards 1 to o refer to a stadtion (ii) while card 7 refers to the stadtion (i).							
	Card 1	Card 2	Card 3	Card 4	Card 5	Card 6	Card 7
Employee A	498 000	117 000	1 000	100 000	166 666	1 000	75 000
Employee B	1 000	142 000	1 000	150 000	166 666	498 000	100 000
Employee C	1 000	241 000	498 000	250 000	166 666	1 000	200 000
Table 2 – The seven cards							

Observe that cards 1 to 6 refer to a situation (iii) while card 7 refers to the situation (i).

If the three *employees* chose the same card, they earn the corresponding amounts; otherwise, it is like card 7 was chosen by all.

Example 1:

Employee A chooses card 7, **employee** B chooses card 5 and **employee** C chooses card 5. The three members of the group do not choose the same card, therefore card 7 is implemented in the group. **Employee** A earns 75 000 ecus, **employee** B earns 100 000 ecus and **employee** C earns 200 000 ecus.

Example 2

Employee A chooses card 1, **employee** B chooses card 1 and **employee** C chooses card 1. The three members of the group choose card 1, therefore this card is implemented. **Employee** A earns 498 000 ecus while **employees** B and C earn 1 000 ecus.

(...)

Communication: treatments (1.0) and (1.2)

When players got the possibility to exchange information (in treatments 1.0 and 1.2) this section was added to the instructions just before section 5 "last details".

The previous sections 1, 2, 3 and 4 are the same as in the instructions of treatments 0.0 and 0.2 for treatments 1.0 and 1.2 respectively. The previous section 5 becomes section 6. New section 5 is as follows.

5) Sending messages

You get the possibility to communicate, still anonymously, with the two other members of your group before the periods 1, 6 and 11. The communication page will be displayed automatically on your screen.

At the top of the screen, you can compose messages using 4 scroll-down menus.

- The first one to indicate the choice or the refusal of a card ("I choose" or "I do not choose").
- The second one to select the card number.
- The third one to choose, among 18 available sentences, the one that corresponds to the card you have chosen or refused.
- The fourth one (non-compulsory) to select another card you want to compare with the chosen one.

Once your message is composed, you click on the button "create the message". Then you have to confirm the creation of your message. All created messages are chronologically added to the list at the bottom of your screen. You can order the messages through the buttons "up" and "down" from the one you consider the most important (at the top of the list) to the least (at the bottom).

Once the list is completed and ordered, you click on the button "send messages" in order to communicate them to the two other members of your group. After having clicked on the button "send messages" you cannot get back to the previous step. When all the members of your group have finished their creation of messages, a new screen appears with all the messages created by the three members of the groups (yours included).

In period 1 you have 8 minutes to compose your messages and order them in the list while in periods 6 and 11 you have 5 minutes. Once the time is over, you cannot modify your list of messages. You must click on the button "send messages".

If you do not want to send messages, all you have to do is to click on the button "quit" (which appears on the screen as long as you do not have created any message). On the next screen it will be indicated at your place (A, B or C) that you have not sent messages.

As for the history of periods, you can consult the history of the exchanged messages during the previous periods by clicking on the button "history of the messages".

Documents de Recherche parus en 2012¹

DR n°2012 - 01 : Abdoul Salam DIALLO, Véronique MEURIOT, Michel TERRAZA « Analyse d'une nouvelle émergence de l'instabilité des prix des matières premières agricoles » DR n°2012 - 02 : Emmanuel DUGUET, Christine Le CLAINCHE « Chronic Illnesses and Injuries: An Evaluation of their Impact on Occupation and Revenues » DR n°2012 - 03 : Ngo Van LONG, Antoine SOUBEYRAN, Raphael SOUBEYRAN « Knowledge Accumulation within an Organization » DR n°2012 - 04 : Véronique MEURIOT « Une analyse comparative de la transmission des prix pour l'orientation des politiques publiques : le cas du riz au Sénégal et au Mali » DR n°2012 - 05 : Daniel SERRA « Un aperçu historique de l'économie expérimentale : des origines aux évolutions récentes » DR n°2012 - 06 : Mohamed CHIKHI, Anne PEGUIN-FEISSOLLE, Michel TERRAZA « Modélisation SEMIFARMA-HYGARCH de la persistance du cours du Dow Jones » DR n°2012 - 07 : Charles FIGUIERES, Fabien PRIEUR, Mabel TIDBALL « Public Infrastructure, non Cooperative Investments and Endogenous Growth » DR n°2012 - 08 : Emmanuel DUGUET, Christine LE CLAINCHE « The Impact of Health Events on Individual Labor Market Histories : the Message from Difference in Differences with Exact Matching » DR n°2012 - 09 : Katrin ERDLENBRUCH, Mabel TIDBALL, Georges ZACCOUR « A Water Agency faced with Quantity-guality Management of a Groundwater Resource » DR n°2012 - 10 : Julia de FRUTOS, Katrin ERDLENBRUCH, Mabel TIDBALL « Shocks in groundwater resource management »

¹ La liste intégrale des Documents de Travail du LAMETA parus depuis 1997 est disponible sur le site internet : http://www.lameta.univ-montp1.fr

DR n°2012 - 11 :	Vanja WESTERBERG, Jette Bredahl JACOBSEN, Robert LIFRAN « The case for offshore wind farms, artificial reefs and sustainable tourism in the French Mediterranean »
DR n°2012 - 12 :	Thierry BLAYAC, Patrice BOUGETTE, Christian MONTET « How Consumer Information Curtails Market Power in the Funeral Industry »
DR n°2012 - 13 :	Sadek MELHEM, Mahmoud MELHEM « Comments on "Re-examining the source of Heteroskedasticity: The paradigm of noisy chaotic models" »
DR n°2012 - 14 :	Raouf BOUCEKKINE, Aude POMMERET, Fabien PRIEUR « Optimal Regime Switching and Threshold Effects : Theory and Application to a Resource Extraction Problem under Irreversibility »
DR n°2012 - 15 :	Raouf BOUCEKKINE, Aude POMMERET, Fabien PRIEUR « On the Timing and Optimality of Capital Controls: Public Expenditures, Debt Dynamics and Welfare »
DR n°2012 - 16 :	Sadek MELHEM, Mahmoud MELHEM « Spéculateurs hétérogènes et volatilité excessive dans le prix du pétrole : une approche dynamique non linéaire »
DR n°2012 - 17 :	Charles FIGUIÈRES, Marc WILLINGER « Regulating ambient pollution when social costs are unknown »
DR n°2012 - 18 :	Christine Le CLAINCHE, Jerome WITTWER « Risky Behaviours and Responsibility-Sensitive Fairness in a Non Life-Threatening Health Case: A European Study »
DR n°2012 - 19 :	Pauline MORNET, Françoise SEYTE, Michel TERRAZA « L'influence du degré d'aversion à l'inégalité du décideur sur sa perception des inégalités intragroupes et intergroupes : une application de l'alpha décomposition aux salaires de la France entre 1995 et 2005 »
DR n°2012 - 20 :	Laure ATHIAS, Raphael SOUBEYRAN « Less Risk, More Effort: Demand Risk Allocation in Incomplete Contracts »
DR n°2012 - 21 :	Lazeni FOFANA, Françoise SEYTE « Modeling Financial contagion: Approach-based on Asymmetric Cointegration »

DR n°2012 - 22 :	Vanja WESTERBERG, Jette BREDAHL JACOBSEN, Robert LIFRAN « The Multi-faceted Nature of Preferences for Offshore Wind Farm Sitting »
DR n°2012 - 23 :	Rachida HENNANI, Michel TERRAZA « Value-at-Risk stressée chaotique d'un portefeuille bancaire »
DR n°2012 - 24 :	Alfred MBAIRADJIM MOUSSA, Jules SADEFO KAMDEM, Michel TERRAZA « Fuzzy risk adjusted performance measures: application to Hedge funds »
DR n°2012 - 25 :	Emmanuel DUGUET / Christine LE CLAINCHE « Une évaluation de l'impact de l'aménagement des conditions de travail sur la reprise du travail après un cancer »
DR n°2012 - 26 :	Christine LE CLAINCHE / Jérôme WITTWER « Preferences for Redistribution : Normative Rationality, Self- Interest and Social Identification »
DR n°2012 - 27 :	Antoine NEBOUT « Sequential decision making without independence : a new conceptual approach »
DR n°2012 - 28 :	Katrin ERDLENBRUCH, Raphael SOUBEYRAN, Mabel TIDBALL, Agnes TOMINI « (Anti-)Coordination Problems with Scarce Water Resources »
DR n°2012 - 29 :	Lisette IBANEZ, Nathalie MOUREAU, Sébastien ROUSSEL « Emotions Incidentes et Comportements Pro-Environnementaux »
DR n°2012 - 30 :	Sébastien ROUSSEL, Jean-Michel SALLES, Léa TARDIEU « Recreation Demand Analysis of the "Sensitive Natural Areas" (Hérault District, France): A Travel Cost Appraisal using Count Data Models»
DR n°2012 - 31 :	Mireille CHIROLEU-ASSOULINE, Jean-Christophe POUDOU, Sébastien ROUSSEL « North / South Contractual Design through the REDD+ Scheme»
DR n°2012 - 32 :	Nicolas QUEROU, Agnès TOMINI « Managing interacting species in unassessed fisheries »

- DR n°2012 33 : Alfred MBAIRADJIM J. SADEFO KAMDEM Arnold F. SHAPIRO Michel TERRAZA « Capital asset pricing model with fuzzy returns and hypothesis testing »
- DR n°2012 34 : Raphaële PREGET Phu NGUYEN-VAN Marc WILLINGER « Who are the Voluntary Leaders? Experimental Evidence from a Sequential Contribution Game »
- DR n°2012 35 : Cécile BAZART Aurélie BONEIN « Reciprocal Relationships in Tax Compliance Decisions»
- DR n°2012 36 : Mathieu Désolé Stefano Farolfi Patrick Rio « How does context influence players' behaviour? Experimental assessment in a 3-player coordination problem »

Contact :

Stéphane MUSSARD : <u>mussard@lameta.univ-montp1.fr</u>

